Author’s response to reviews

Title: Assessment of variation in the Alberta Context Tool: The contribution of unit level contextual factors and practice specialty in Canadian pediatric acute care settings

Authors:

Carole A Estabrooks (carole.estabrooks@ualberta.ca)
Janet E Squires (jasquires@ohri.ca)
Alison M Hutchinson (alison.hutchinson@deakin.edu.au)
Shannon Scott (shannon.scott@ualberta.ca)
Greta G Cummings (greta.cummings@ualberta.ca)
Sung Hyun Kang (sunghyun.kang@nurs.ualberta.ca)
William K Midodzi (wmidodzi@med.mun.ca)
Bonnie Stevens (bonnie.stevens@sickkids.ca)

Version: 5 Date: 30 September 2011

Author’s response to reviews:

We have corrected the manuscript for some minor typographical errors.